Author’s response to reviews

Title: Long-term survival of cancer patients compared to heart failure and stroke: A systematic review.

Authors:

Vasileios Askoxylakis (vasileios.askoxylakis@med.uni-heidelberg.de)
Christian Thieke (c.thieke@dkfz.de)
Sven T Pleger (sven.pleger@med.uni-heidelberg.de)
Patrick Most (patrick.most@jefferson.edu)
Judith Tanner (judith.tanner@med.uni-heidelberg.de)
Katja Lindel (katja.lindel@med.uni-heidelberg.de)
Hugo A Katus (hugo.katus@med.uni-heidelberg.de)
Juergen Debus (juergen.debus@med.uni-heidelberg.de)
Marc Bischof (marc.bischof@med.uni-heidelberg.de)

Version: 3 Date: 18 March 2010

Author’s response to reviews:

Dear Sir or Madam,

we are pleased to offer a revised version of our manuscript "Long-term survival of cancer patients compared to heart failure and stroke: A systematic review". The manuscript has been revised as suggested. The reference 47 has been moved to position 24 of the reference list and is cited in the manuscript (page 9, Grivaux et al...was 7 months). Figure 1 is cropped as closely as possible, as suggested. The manuscript has been checked for typographical errors.

We are looking forward to hearing from you,

Sincerely yours,
Vasileios Askoxylakis, MD